Author's response to reviews

Title: Social adaptation in Chinese Williams syndrome

Authors:

chai ji (jichai1213@163.com)
dan yao (158965672@qq.com)
weijun chen (chenwj0575@hotmail.com)
ingyan li (lmy524@163.com)
zhengyan zhao (chebk@zju.edu.cn)

Version: 2 Date: 10 September 2013

Author's response to reviews: see over
Dear Editors,

We are submitting a manuscript entitled “Social adaptation in Chinese Williams syndrome” for consideration by BMC pediatrics. This paper reports the social ability and behavioral development in WS patients in China. This topic is important in neurodevelopmental study in WS patients, since it may help us enhance understanding of these patients’ social ability development and identify further targets of intervention to improve healthy social development. However, to our knowledge it hasn’t been well resolved in china before.

All authors are responsible for the reported research and participated in the concept and design, analysis and interpretation of data, drafting or revising of the manuscript. All the authors listed have approved the manuscript that is enclosed.

This work was supported by grants from Zhejiang Province innovation team for early screening and intervention of birth defects. (Grant No. 2010R50045). The content of this article is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health. No author has a conflict of interest to declare.

Thank you for your kind consideration.

Yours truly,

Zhengyan Zhao, MD
Professor
Department of Child Health Care
Children’s Hospital Zhejiang University School of Medicine
57 Zhugan Xiang Road
Hangzhou, Zhejiang 310003